Author’s response to reviews

Title: A Systematic Review on the Prevalence of Dietary Supplement Use by Military Personnel

Authors:

Joseph J Knapik (joseph.j.knapik.ctr@mail.mil)
Ryan A Steelman (ryan.a.steelman.ctr@mail.mil)
Sally S Hoedebecke (shfdressage@gmail.mil)
Emily K Farina (emily.k.farina.ctr@mail.mil)
Krista G Austin (krista.g.austin.ctr@mail.mil)
Harris R Lieberman (harris.r.lieberman.civ@mail.mil)

Version: 2 Date: 23 October 2013

Author's response to reviews: see over
Dear Sir or Madam:

Please consider our manuscript entitled “A Systematic Review of the Prevalence of Dietary Supplement Use by Military Personnel” for publication in the journal BMC Complementary and Alternative Medicine. This paper is the first to systematically review, and perform a meta-analyses on, the prevalence of dietary supplement uses in the military services. The paper also examines the range of dietary supplements used by military personnel.

This is an original work and is not under consideration by any other journal. No author has a financial or other relationship in this paper that might be perceived as leading to a conflict of interest (i.e., affecting author objectivity). The manuscript has been read by and approved by all authors.

We recommend the following individuals as appropriate to peer review this manuscript: Dr Sonya Cable (sonya.j.cable.mil@mail.mil), Dr Nicole Cassler (nicole.m.cassler.mil@health.mil), or Dr Robin Toblin (robin.l.toblin.mil@mail.mil). All have practical and research experience with dietary supplements in military populations.

Please direct correspondence regarding this manuscript to me as follows:
Dr Joseph Knapik
US Army Research Institute of Environmental Medicine
45 Kansas Street
Natick MA 01760
My phone number is (443) 752-3350 and my e-mail address is joseph.j.knapik.ctr@mail.mil.

Joseph J Knapik, ScD
Research Physiologist